Author’s response to reviews

Title: Birth ball for pregnant women in labour research protocol: a multi-centre randomised controlled trial

Authors:
May Pui Shan YEUNG (may.yeung@cuhk.edu.hk)
Katrina Wai Kay TSANG (cuhkfm@cuhk.edu.hk)
Benjamin Hon Kei YIP (benyip@cuhk.edu.hk)
Wing Hung TAM (tamwh@cuhk.edu.hk)
Wan Yim IP (ip2013@cuhk.edu.hk)
Florence Wai Lei HAU (hwl081@ha.org.hk)
Margaret Kit Wah WONG (wk387@ha.org.hk)
Judy Wai Ying NG (nwy563@ha.org.hk)
Sau Ha LIU (liush2@ha.org.hk)
Sophia Shu Wing CHAN (sophiachan@cuhk.edu.hk)
Chi Kin LAW (c.law1@griffith.edu.au)
Samuel Yeung Shan WONG (yeungshanwong@cuhk.edu.hk)

Version: 6 Date: 17 Apr 2019

Author’s response to reviews:

Dear Editor,

Thank you for your comment and email on the 9 April 2019. Here is the point-by-point response to the letter / email.

The questionnaires I do not wish to publish are removed from the supplementary files.
The references of questionnaires developed by other authors in the Methods section are added in the manuscript. The reference numbers are added to “Methods” section, “Outcome measurements”, “i. Primary outcomes” and “ii. Secondary outcomes” on p.7 line 55 and p.8 line 1 and line 11 of the main text. The new references are numbers 21 to 26 of p.14.

Thank you.

Best regards,

May